Author's response to reviews

Title: Consistency between stated and revealed preferences: a Discrete Choice Experiment and a behavioral experiment on vaccination behavior compared.

Authors:

Mattij S Lambooij (mattijs.lambooij@rivm.nl)
Irene A Harmsen (irene.harmsen@rivm.nl)
Jorien Veldwijk (jorien.veldwijk@rivm.nl)
Hester de Melker (hester.de.melker@rivm.nl)
Liesbeth Mollema (liesbeth.mollema@rivm.nl)
Yolanda WM van Weert (Yolanda.van.weert@rivm.nl)
Ardine AG de Wit (ardine.de.wit@rivm.nl)

Version: 3
Date: 14 February 2014

Author's response to reviews: see over
Editorial team of BMC Medical Research Methodology  
BioMed Central  
236 Gray’s Inn Road  
London WC1X 8HB  
United Kingdom  

Februari 14, 2014  

Dear professor Pala,  

Hereby I resubmit the paper ‘Consistency between stated and revealed preferences: a Discrete Choice Experiment and a behavioral experiment on vaccination behavior compared.’ For publication in BMC Medical Research Methodology.  

We added a document with a point by point reply to the comments of the reviewers. We wish to thank the reviewers for their constructive comments. We have made several clarifications in the text, and we were able to improve our analyses with their comments. We believe it improved the paper substantially. If the paper gets accepted for publication, we would like to acknowledge the contribution of the reviewers.  

Please contact me if you have any further questions.  

Sincerely yours,  
on behalf of all authors  

Mattijs Lambooij, PhD.  
Dept. Quality of Care and Health Economics  
National Institute of Public Health and the Environment  
PO box 1  
3720 BA, Blihoven  
The Netherlands  
Mattijs.lambooij@rivm.nl